Author’s response to reviews

Title: Sheng-ji Hua-yu Formula Promotes Diabetic Wound Healing of Re-epithelization via Activin/Follistatin Regulation

Authors:

Le Kuai (mjbubu@qq.com)
Fu-lun Li (drlifulun@163.com)
Jing-ting Zhang (zjt08310@163.com)
Bin Li (18930568129@163.com)
Yu Deng (yudeng1211@126.com)
Shun Xu (Shunxu1211@126.com)
Min-feng Wu (996392167@qq.com)
Dong-jie Guo (guodongjie123@126.com)
Yu Chen (cy11235@126.com)
Ren-jie Wu (fuchuwrj@163.com)
Xun-zhe Xu (xxz@163.com)
Xing-qiang Zhao (315368004@163.com)
Hua Nian (drnianhua@126.com)

Version: 2 Date: 27 Dec 2017

Author’s response to reviews:

Dear Editor,

Thank you very much for your letter and advice on our manuscript. Those comments are very helpful for revising our paper. I have carefully made a revision and the responses are as following. We are also submitting a clean version of the manuscript that do not have tracked changes or highlighting. Once again, thank you very much for your comments and suggestions. We hope that the revision is acceptable and look forward to hearing from you soon.
Editor Comments:

1. Please provide a figure legend for "raphical abstract BMC 6.1.tif", and upload it as a regular figure. Please include the figure legend with your list of figure legends.

   "raphical abstract BMC 6.1.tif" has been uploaded as figure1, and we have included the figure legend with your list of figure legends.

2. Please change the sentence "The detection of SJHY Formula was analyzed by LC-MS" as "The SJHY Formula was analyzed by LC-MS".

   "The detection of SJHY Formula was analyzed by LC-MS" has been changed as "The SJHY Formula was analyzed by LC-MS".

Sincerely
Fu-lun Li
Professor
E-mail: drlifulun@163.com